Author's response to reviews

Title: Physical inactivity is associated with chronic musculoskeletal complaints 11 years later: Results from the Nord-Trøndelag Health Study

Authors:

Helene S Holth (helenesu@stud.ntnu.no)
Hanne KB Werpen (werpen@stud.ntnu.no)
John-Anker Zwart (j.a.zwart@medisin.uio.no)
Knut Hagen (knut.hagen@ntnu.no)

Version: 5 Date: 28 November 2008

Author's response to reviews:

Dear BioMed Central Editorial Team

Re: “Physical activity is associated with chronic musculoskeletal complaints 11 years later: results from the Nord-Trøndelag Health Study”

The following formatting changes are performed:

1. The information in the former Table 1 is included within the manuscript text on page 6 and 7. The former Tables 2-4 have been changed to Tables 1-3.
2. Email address: The suggested format is used.

On behalf of all authors;
Sincerely yours,
Knut Hagen, MD, PhD
Norwegian National Headache Centre
Department of Neurology
St. Olav’s Hospital
7006 Trondheim
Norway
E-mail: knut.hagen@ntnu.no